Author's response to reviews

Title: Predictors of Effectiveness of Early Intervention on Children with Intellectual Disability

Authors:

Der-Chung Lai (cych02315@gmail.com)
Chung-Hsin Chiang (chchiang@nccu.edu.tw)
Yu-Ming Hou (02151@cych.org.tw)
Jiun-Horng Liu (eternityliu2001@yahoo.com.tw)
Shu-Fen Yao (03500@cych.org.tw)
How-Ran Guo (hrguo@mail.ncku.edu.tw)
Yen-Cheng Tseng (yvonneych@yahoo.com)

Version: 2
Date: 31 October 2013

Author's response to reviews:

Dear Editor,

Thank you for your instructions on the revision of our manuscript with the title of “Predictors of Effectiveness of Early Intervention on Children with Intellectual disability.” The responses to your requests are as following:

(1) Ethics statement: In the original manuscript, we put “This study was approved by the Institutional Review Board of the hospital.” at the end of the “Data collection” section in the “Method.” In response to the request, we changed the statement to “This study was approved by the Institutional Review Board of the Chia-Yi Christian Hospital.” in the revised manuscript.

(2) Competing Interest: As requested, we added a “Competing Interests” section in the revised manuscript, which include the statement “The authors declare that they have no competing interests.”

(3) Authors’ contributions: As requested, we added an “Authors’ Contributions” section after the Competing Interests section in the revised manuscript.

(4) Consent: Because all the data used in this study were from the medical records at the hospital and were analyzed anonymously, the Institutional Review Board of the Chia-Yi Christian Hospital did not require consent from each participant. This is a general practice in Taiwan.

(5) References: As requested, we assigned individual number to each reference in the revised manuscript.

If you have any further requests or questions, please do not hesitate to contact me.
Sincerely,

Yen-Cheng Tseng